Author’s response to reviews

Title: Current Approaches in the Surgical Treatment of Liver Hydatid Disease

Authors:
mehmet bayrak (drmehmetbayrak@hotmail.com)
yasemin altintas (yasminoter@yahoo.com)

Version: 5 Date: 29 Jun 2019

Author’s response to reviews:

Answer to Editor Comments:

1. Reformat as a research article

Please edit your manuscript based on the below criteria and resubmit as 'article type: research article'.

   Background: the context and purpose of the study

   Methods: how the study was performed and statistical tests used

   Results: the main findings

   Conclusions: brief summary and potential implications

-We edited our article according to the reformat as a research article. We changed reformat as a research article.
-We changed ‘Material and Methods’ as ‘Methods’ in the main text.

2. Authors Contributions
-- Please represent authors' names using their full initials, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

BMC Surgery operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

-We represent authors' names using their full initials, in the Authors’ Contributions section
- We included ‘a statement’ in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

3. Consent

Please clarify written informed consent was obtained from parents for any minors (below 16).

-We clarified that written informed consent was obtained from parents for two minors (below 16).

4. Abbreviations

- If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided just before the Declarations.

-We checked abbreviations used in the text and provided just before the Declarations.
Currently, there is no additional data attached as an additional file.

In the "Availability of Data and Materials" section, please detail where the raw data supporting your findings can be found (including information found in any supplementary files). If the raw data is publicly available or can be requested, please state that this is the case, and where the data can be found/requested from. Alternatively, if you do not wish to/cannot share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason.

- Currently, there is no additional data attached as an additional file.
- The datasets used and/or analyzed during the current study are attached as supplementary files.

- Thank you for your revision.

Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished-writing-resources). These cover common mistakes that occur when writing in English.
Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements, this may cause a delay while this is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Surgery must contain a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcsurg.biomedcentral.com/).

Where a mandatory Declarations section is not relevant to your study design or article type, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).

Declarations

Ethics approval and consent to participate

Consent to publish

Availability of data and materials

Competing interests

Funding
Authors' Contributions

Acknowledgements

-We checked our article as editorial policies.